RESEARCH

Open Access

Gestational weight gain during the second and third trimesters and adverse pregnancy outcomes, results from a prospective pregnancy cohort in urban Tanzania

Jiaxi Yang^{1,2,3*}, Molin Wang^{1,4}, Deirdre K. Tobias^{5,6}, Janet W. Rich-Edwards^{1,7}, Anne Marie Darling⁸, Ajibola I. Abioye⁵, Andrea B. Pembe⁹, Isabel Madzorera⁸ and Wafaie W. Fawzi^{1,5,8}

Abstract

Background: Appropriate gestational weight gain (GWG) is important for optimal pregnancy outcomes. This study prospectively evaluated the associations between GWG during the second and third trimesters of pregnancy and adverse pregnancy outcomes in an urban Tanzanian pregnancy cohort.

Methods: We used data from a randomized clinical trial conducted among pregnant women recruited by 27 weeks of gestation in Dar es Salaam, Tanzania (N = 1230). Women's gestational weight was measured at baseline and at monthly antenatal visits. Weekly GWG rate during the second and third trimesters was calculated and characterized as inadequate, adequate, or excessive, in conjunction with measured or imputed early-pregnancy BMI status according to the 2009 Institute of Medicine (IOM) GWG guidelines. We used multivariable Poisson regression with a sandwich variance estimator to calculate risk ratios (RR) for associations of GWG with low birth weight, preterm birth, small for gestational age (SGA), and large for gestational age (LGA). Degree of appropriate GWG defined using additional metrics (i.e., percentage of adequacy, z-score) and potential effect modification by maternal BMI were additionally evaluated.

Results: According to the IOM guidelines, 517 (42.0%), 270 (22.0%), and 443 (36.0%) women were characterized as having inadequate, adequate, and excessive GWG, respectively. Overall, compared to women with adequate GWG, women with inadequate GWG had a lower risk of LGA births (RR = 0.54, 95% CI: 0.36–0.80) and a higher risk of SGA births (RR = 1.32, 95% CI: 0.95–1.81). Women with inadequate GWG as defined by percentage of GWG adequacy had a higher risk of LGW (OR = 1.93, 95% CI: 1.03–3.63). In stratified analyses by early-pregnancy BMI, excessive GWG among women with normal BMI was associated with a higher risk of preterm birth (RR = 1.59, 95% CI: 1.03–2.44).

Conclusions: A comparatively high percentage of excessive GWG was observed among healthy pregnant women in Tanzania. Both inadequate and excessive GWGs were associated with elevated risks of poor pregnancy outcomes.

*Correspondence: jiaxiyang@nus.edu.sg

¹ Department of Epidemiology, Harvard T. H. Chan School of Public Health, 677 Huntington Ave, Boston, MA 02115, USA Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

Future studies among diverse SSA populations are warranted to confirm our findings, and clinical recommendations on optimal GWG should be developed to promote healthy GWG in SSA settings.

Trial registration: This trial was registered as "Prenatal Iron Supplements: Safety and Efficacy in Tanzania" (NCT01119612; http://clinicaltrials.gov/show/NCT01119612).

Keywords: Gestational weight gain, Adverse birth outcomes, Tanzania, Institute of Medicine (U.S.)

Plain language summary

Pregnancy is a critical lifetime event for both mother and the offspring, with implications in short-term and long-term health consequences. Gestational weight gain (GWG) is an important modifiable factor for pregnancy outcomes related to infant body size and weight and prematurity. Countries in sub-Saharan Africa (SSA) have long had poor rates of insufficient GWG and pregnancy complications associated with insufficient GWG. Nevertheless, some SSA countries are experiencing economic transitions accompanied with changes in lifestyle and nutrition, which might impact pregnancy experiences, including GWG and pregnancy outcomes. This study aimed to characterize recent GWG patterns and the associations of both inadequate and excessive GWG with adverse pregnancy outcomes, using an urban pregnancy cohort in Tanzania. This study found that 42.0%. 22.0%, and 36.0% of women had insufficient, adequate, and excessive GWG, respectively. Insufficient GWG was associated with higher risks of small infant size and low infant body weight, and excessive GWG was associated with higher risk of preterm birth, particularly among women with body mass index 18.5–25.0 kg/m². Results from the present study highlight that both insufficient and excessive GWG are of potential public health concerns in urban centers of SSA, concerning upward trends in obesity and possibly obesity-related pregnancy consequences. Local public health practitioners should continue to

advocate longitudinal GWG monitoring and care among African pregnant women, and optimal GWG with feasible and effective clinical guidelines should be developed to prevent both over- and under-gaining of maternal weight during pregnancy.

Background

Pregnancy is a key health-related event for both the mother and the offspring. Women who developed pregnancy complications have higher risk of future pregnancy complications [1] and poor long-term health outcomes, including obesity [2], and metabolic [3] and cardiovascular diseases [4, 5]. Addressing poor birth outcomes is a critical step in improving child survival at a global scale [6]. Adverse birth outcomes, including prematurity and inappropriate intrauterine growth, have been associated with neonatal complications, higher neonatal and infant mortality [7], as well as life-long health and developmental problems in the offspring [8]. Therefore, identifying and intervening on factors associated with adverse pregnancy outcomes remains a critical means of addressing and preventing pregnancy-related complications, particularly for low- and middle-income countries where the rates of pregnancy complications remain high [9, 10].

Gestational weight gain (GWG) is one of the key modifiable factors associated with birth outcomes [11, 12]. Components of GWG include body composition of the mother, weight of the fetus, placenta, and amniotic fluid [13]. GWG reflects the health status of the mother and the growth of the fetus [13]. Most of the maternal weight gain takes place after the first trimester of a pregnancy [14, 15], with GWG in the second and third trimesters contributing to about 80–90% of the total GWG [13]. Prospective studies conducted in Caucasian or Asian populations have supported the associations between GWG in the second and the third trimesters and pregnancy outcomes related to infant body weight and size and prematurity [13, 15, 16].

Distinct GWG patterns have been observed across world regions, with overall higher proportions of suboptimal GWG especially in populations from lowincome countries such as in sub-Saharan Africa (SSA), a region that has long had high rates of poor birth outcomes [7, 17-19]. Nonetheless, patterns of GWG across different African contexts are changing and may have become more heterogenous than they were in the past. Some SSA countries, such as South Africa, Kenya, and Tanzania, have recently undergone transition from low-income to middle-income status, accompanied with changes in lifestyles, including better food security, dietary transitions, and reduced physical activity [20, 21]. These changes may have led to changes in maternal diets prior to and during pregnancy, affecting GWG patterns and the overall pregnancy experience for women in these regions [22]. For example, a recent study conducted in South Africa reported that the percentage of excessive GWG was as high as 55%, which was even higher than the rates in some developed countries [23].

While the associations between GWG and pregnancy outcomes have been well examined in other countries, evidence from SSA remains inadequate. Several studies in SSA examined these associations, but they were largely limited by retrospective or cross-sectional designs, insufficient measures of pregnancy weight, and inadequate adjustment for key confounders, including pre-pregnancy BMI [17]. Of particular importance, the noted nutrition transition underway in urban centers of SSA with rising rates of overweight and obesity is likely not reflected in earlier studies. Therefore, we sought to prospectively examine the associations between GWG in the second and third trimesters and adverse birth outcomes in a healthy pregnancy cohort in Tanzania.

Methods

Study population

We used data from a randomized clinical trial conducted in urban Tanzania. Details of this study have been described elsewhere [24, 25]. Briefly, from September 2010 to October 2012, a randomized trial on iron supplements was conducted in Dar es Salaam, Tanzania. Participants were screened and enrolled at antenatal care clinics. Women were eligible if they were iron-replete, non-anemic, HIV-uninfected, primigravidae or secundigravidae, and were recruited at or before 27 weeks of gestation. Baseline gestational age (weeks) was estimated based on the reported timing of the last menstrual period (LMP). The study enrolled 1500 pregnant women who were subsequently randomized to receive a daily oral dose of either 60 mg of iron or placebo from the time of enrollment until delivery.

At baseline, women completed a sociodemographic and reproductive health questionnaire, as well as a full clinical examination. They were subsequently followed at monthly antenatal visits and at time of delivery. For our study, we excluded participants with unknown gestational age at delivery (n=22), unknown delivery outcomes (n=15), or twin babies (n=27). Since GWG in the second and third trimesters was the main exposure of interest, we further excluded women with only one weight measure during that time window (n=206), leaving us with a final study sample of 1230 participants.

Assessment and characterization of GWG

Study participants' weight at baseline and at monthly follow-up visits was measured by trained study nurses using a calibrated weight scale. Pre-pregnancy BMI has been suggested as an important covariate for the association between GWG and pregnancy outcomes [13]. However, information on pre-pregnancy weight was not collected in the original trial study. Further, given the distribution of the baseline gestational age at enrollment, only 196 out of 1230 participants were enrolled during the first trimester, of which the majority were enrolled in the late first trimester (interquartile range of gestational age at enrollment among participants enrolled in the first trimester: 10-13 weeks). We therefore imputed BMI at the end of the first trimester (14 weeks of gestation) for covariate adjustment and stratification. Based on the repeated weight measurements during pregnancy, we fit mixedeffects models with polynomial terms of gestational age (weeks) and imputed individual-specific weight at 14 weeks of gestation; statistical results suggested good imputation performance (mean absolute error: 1.95 kg, concordance rate of categorical BMI among women with available first-trimester weight: 89.0%). Details on the statistical methods and the imputation results can be found elsewhere [25]. Based on the imputed weight at the end of the first trimester and the height measured at baseline, the corresponding BMI status at the end of the first trimester was derived (underweight if BMI < 18.5 kg/ m², normal if 18.5 kg/m² \leq BMI < 25 kg/m², overweight if 25 kg/m² \leq BMI < 30 kg/m², and obese if BMI \geq 30 kg/ m^{2}).

The main exposure of interest was GWG during the second and third trimesters. We defined degree of appropriate GWG based on the 2009 Institute of Medicine (IOM) guidelines [13]. The IOM guidelines provided recommended ranges for total weight gain and the rate of weight gain during the second and third trimesters, based on pre-pregnancy BMI status. Weekly rate of GWG during the second and third trimesters (kg/week) was derived by calculating the difference between the first measured weight in the second trimester and the last measured weight before delivery and dividing that by the number of weeks between the two measures. For each given participant, based on the calculated weekly rate of GWG, the BMI status at the end of the first trimester, and the IOM recommended GWG range (0.44-0.58 kg/week for underweight, 0.35-0.50 kg/week for normal weight, 0.23-0.33 kg/week for overweight, and 0.17-0.27 kg/ week for obese), GWG was characterized as inadequate (weekly rate of GWG below the recommended range), adequate (weekly rate of GWG within the recommended range), or excessive (weekly rate of GWG above the recommended range) [13]. We made assumptions that weight gain during the first trimester was minimal and that women stayed in the same BMI category from the start of the pregnancy until the end of the first trimester [13].

We additionally characterized GWG using other metrics, including percentage of GWG adequacy (i.e.

percentage method) [26] and GWG z-score (i.e., z-score method) based on the INTERGROWTH-21st standard [27]. Building upon the IOM guidelines which grouped the extent of GWG into three categories (i.e., inadequate, adequate, and excessive GWG), the percentage method provided a percentage value to further quantify the amount of GWG relative to the guidelines with accounting for the pregnancy duration. Details on this method has been described elsewhere [26]. Briefly, percentage adequacy of GWG was calculated as the ratio of observed weight gain (kg) and expected weight gain (kg) during pregnancy. The original formula was given as follows: percent adequacy=observed weight gain during pregnancy / [expected first trimester weight gain+((week at the last weight measure - 13)*expected weekly GWG rate in the second and third trimesters)]. Given the research question of our study, we modified the formula by restricting the time period to the second and third trimesters instead of the entire pregnancy. Consistent with Adu-Afarwuah et al., we derived BMI-specific percentage cutoffs based on the IOM cutoffs and classified the GWG into three groups based on the calculated percent adequacy: inadequate, adequate, and excessive, respectively [26].

We further constructed a GWG z-score (in unit of standard deviation) for participants with a normal BMI $(18.5 \text{ kg/m}^2 \le \text{BMI} < 25 \text{ kg/m}^2)$ at the end of the first trimester, using a standard reference chart developed by the INTERGROWTH-21st consortium [27]. Briefly, by applying the reference chart, a gestational age-specific GWG z-score can be derived based on the total weight (kg) gained up to a given gestational age. Since GWG was likely to follow a non-linear trajectory over the course of pregnancy, a gestational age-specific z-score could account for the natural correlation between a longer pregnancy duration and a higher rate of GWG [28], which, if unaddressed, could bias the association between GWG and gestational age-related outcome (e.g., prematurity). For our analysis, total weight gain in the second and third trimesters and gestational age at the last weight measure were used to derive the z-score. Given the potential nonlinearity of the z-score with respect to risks of pregnancy outcomes and the distribution of the z-scores in our sample (only one participant had z-score>2 units), we classified participants into one of the two following groups: inadequate GWG if z-score < -2 units (2.3th percentile), adequate GWG if z-score within ± 2 units (between 2.3th and 97.7th percentile).

Outcome assessment

At the time of delivery, on-site midwives recorded participants' pregnancy outcomes. Data on gestational age at delivery (weeks), delivery outcome if known (miscarriage [n=1], stillbirth [n=47], and live birth [n=1182]), infant sex, and infant birth weight (kg) were determined. We derived the following outcome variables for pregnancies resulting in live births: low birth weight (LBW, birthweight < 2.5 kg), preterm birth (gestational age at delivery < 37 weeks), small for gestational age and large for gestational age (SGA and LGA, gender-specific birth weight below the 10th percentile and above the 90th percentile respectively for babies of the same gestational age according to the INTERGROWTH-21st reference) [29]. Although we did not have information on the type of preterm birth (i.e., spontaneous, medically induced), we considered most of the preterm cases as spontaneous, based on conversations with on-site research staff and medically induced preterm birth being relatively uncommon in Tanzania.

Statistical analysis

In the main analyses, GWG during the second and third trimesters according to the IOM recommendations was evaluated with respect to adverse birth outcomes. GWG with three levels defined by the IOM guidelines (i.e., inadequate, adequate, and excessive GWG) was modeled as a categorical variable, and the group with adequate GWG was set as the reference group. The following dichotomous outcomes were examined: LBW, preterm birth, SGA, and LGA. We used multivariable Poisson regression with a sandwich variance estimator to estimate risk ratio (RR) and 95% confidence interval (CI) [30]. We adjusted for covariates hypothesized a priori as potential confounders in the analyses, including age, baseline gestational age, gestational age at delivery, measured or imputed BMI at 14 weeks of gestation, primigravida status, treatment status, marital status, education, occupation, and history of prior complications (history of cardiovascular disease, high blood pressure, diabetes, or weight loss in previous year, or ever had a LBW baby or non-live birth among non-primigravida).

Given the evidence of the heterogeneity by pre-pregnancy BMI status for the associations of interest [13], we further stratified the analyses by BMI status at the end of the first trimester. Due to the limited sample size, we did not examine these associations among underweight women (n=72) and examined the questions among women with normal BMI (18.5 kg/m² \leq BMI < 25 kg/m²; n=756) and overweight or obese women (BMI \geq 25 kg/m²; n=402 [295 and 107 for overweight and obese, respectively]), separately. Heterogeneity was evaluated by the statistical significance of the cross-product term between categorical GWG and BMI status in the analysis sample excluding underweight women.

In the sensitivity analyses, we additionally examined appropriate GWG and birth outcomes, using the percentage and the z-score methods. For the z-score method, since the INTERGROWTH- 21^{st} GWG reference chart is currently available only for women with normal pre-pregnancy BMI, we restricted the analyses to participants with normal BMI at the end of the first trimester (n=755). All analyses were conducted using SAS statistical software (version 9.4; SAS Institute Inc, Cary, NC, USA). All statistical tests were 2-sided, with *p*-value less than 0.05 considered as statistically significant.

Results

Our analysis included 1,230 women, with a mean baseline gestational age of 17.9 weeks (Table 1). According to the IOM recommendations, 517 (42.0%), 270 (22.0%), and 443 (36.0%) women had inadequate, adequate, and excessive GWG, respectively. Compared to women with adequate GWG, women with inadequate GWG were more likely to be unemployed and report a prior history of complications. Women with excessive GWG were more likely to be primigravida and have a higher early-pregnancy BMI, better educational status, and skilled occupation (Table 1).

GWG-related characteristics and pregnancy outcomes for the entire sample and by the IOM-defined GWGs were summarized in Table 2. Classifications of appropriate GWG by the IOM guidelines and the percentage method were overall consistent: 22.0% were classified as adequate GWG using the IOM method, compared with 30.6% using the percentage method. Yet, compared to the GWG defined by the IOM recommendations, the percentage method was slightly more conservative, as it classified more women having inadequate or adequate GWG and fewer women having excessive GWG. Among women with normal

Table 1 Study population baseline characteristics overall and by status of GWG according to the 2009 IOM guidelines

Baseline characteristics	Entire dataset (n = 1230)	2009 IOM GWG guidelines ^a			
		Inadequate GWG (n = 517, 42.0%)	Adequate GWG (n = 270, 22%)	Excessive GWG (n = 443, 36%)	
Mean (SD)					
Age at baseline (years)	24.1 (4.2)	23.9 (4.4)	24.1 (3.8)	24.2 (4.2)	
Weight at baseline	59.9 (11.7)	59.1 (12.0)	58.3 (10.5)	61.8 (11.8)	
Height at baseline (cm)	156.2 (6.1)	156.0 (6.3)	155.8 (5.8)	156.7 (5.9)	
Gestational age at baseline (weeks)	17.9 (4.3)	18.1 (4.5)	17.5 (4.2)	18.0 (4.1)	
Total number of antenatal visits ^b	5 (2–9)	5 (2–8)	5 (2–8)	5 (2–9)	
n (%)					
Treatment (iron supplement)	601 (48.9)	247 (47.8)	137 (50.7)	221 (49.9)	
Primigravida	706 (57.4)	285 (55.1)	150 (55.6)	271 (61.2)	
Marital status (married)	979 (79.6)	420 (81.2)	219 (81.1)	345 (77.9)	
BMI at baseline (kg/m²)	24.5 (4.6)	24.3 (4.7)	24.1 (4.2)	25.1 (4.5)	
BMI at 14 weeks of gestation (kg/m ²)	24.0 (4.3) ^d	23.5 (4.4)	23.5 (4.0)	24.9 (4.4)	
Education status					
0-4 years	61 (5.0)	25 (4.8)	14 (5.2)	22 (5.0)	
5–7 years	645 (52.4)	286 (55.3)	147 (54.4)	212 (47.9)	
8–11 years	343 (27.9)	136 (26.3)	75 (27.8)	132 (29.8)	
\geq 12 years	181 (14.7)	70 (13.5)	34 (12.6)	77 (17.4)	
Occupation status					
Unemployed	619 (50.3)	274 (53.0)	134 (49.6)	211 (47.6)	
Unskilled or informal	381 (31.0)	154 (29.8)	89 (33.0)	138 (31.2)	
Skilled	230 (18.7)	89 (17.2)	47 (17.4)	94 (21.2)	
History of prior complications ^c	1–91 (15.5)	49 (18.2)	78 (15.1)	64 (14.5)	

Institute of Medicine (IOM), gestational weight gain (GWG), standard deviation (SD), body mass index (BMI)

^a The IOM provided recommended ranges of weekly GWG rate during the 2nd and 3rd trimesters (kg/week) by pre-pregnancy BMI status: 0.44––0.58 kg/week for underweight, 0.35–0.50 kg/week for normal weight, 0.23–0.33 kg/week for overweight, and 0.17–0.27 kg/week for obese. BMI categories were defined according to the WHO standard BMI guidelines

^b Median (interquartile range) were presented

^c History of prior complications was defined as reporting any of the following: cardiovascular disease, high blood pressure, diabetes, weight loss in previous year, ever having a low-birth-weight baby or non-live birth (fetal death, abortion, miscarriage, ectopic pregnancy) among non-primigravida

^d Imputed gestational weight at 14 weeks of gestation

Table 2 Summar	ry characteristics of GWG and	pregnancy outcomes in the study	y population overall and by GWG status

Outcomes	Entire dataset	2009 IOM GWG guidelines ^a			
	(n=1230)	Inadequate GWG (n = 517, 42.0%)	Adequate GWG (n = 270, 22.0%)	Excessive GWG (n = 443, 36.0%)	
GWG-related outcomes					
Weight gain (kg), mean (SD)	6.3 (4.9)	2.8 (3.4)	6.9 (2.8)	10.1 (4.4)	
Rate of weight gain in 2nd–3rd trimester (kg/week), mean (SD)	0.38 (0.32)	0.14 (0.21)	0.39 (0.09)	0.65 (0.28)	
GWG adequacy, n (%) ^b					
Inadequate GWG	553 (45.0)	456 (88.2)	82 (30.4)	15 (3.4)	
Adequate GWG	377 (30.6)	48 (9.3)	174 (64.4)	155 (35.0)	
Excessive GWG	300 (24.4)	13 (2.5)	14 (5.2)	273 (61.6)	
Adverse pregnancy outcomes					
Gestational age at delivery (weeks), mean (SD)	39.5 (3.3)	39.7 (3.6)	39.6 (2.7)	39.0 (3.3)	
Infant birth weight (kg), mean (SD)	3.1 (0.5)	3.1 (0.5)	3.2 (0.5)	3.2 (0.6)	
Low birth weight (< 2.5 kg), n (%)	92 (7.5)	40 (7.7)	15 (5.6)	37 (8.4)	
Preterm birth (< 37 weeks), n (%)	195 (15.9)	76 (14.7)	41 (15.2)	78 (17.6)	
SGA, n (%) ^c	199 (16.2)	101 (19.5)	40 (14.8)	58 (13.1)	
LGA, n (%) ^c	134 (10.9)	36 (7.0)	38 (14.0)	60 (13.5)	
Stillbirth, n (%) ^d	47 (3.8)	12 (2.3)	9 (3.3)	26 (5.9)	

Institute of Medicine (IOM), gestational weight gain (GWG), standard deviation (SD), small for gestational age (SGA), large for gestational age (LGA)

^a The IOM provided recommended ranges of weekly GWG rate during the 2nd and 3rd trimesters (kg/week) by pre-pregnancy BMI status: 0.44–0.58 kg/week for underweight, 0.35–0.50 kg/week for normal weight, 0.23–0.33 kg/week for overweight, and 0.17–0.27 kg/week for obese. BMI categories were defined according to the WHO standard BMI guidelines

^b GWG adequacy was calculated based on the method described in Adu-Afarwuah, Seth, et al. "Maternal supplementation with small-quantity lipid-based nutrient supplements compared with multiple micronutrients, but not with iron and folic acid, reduces the prevalence of low gestational weight gain in semi-urban Ghana: a randomized controlled trial." The Journal of nutrition 147.4 (2017): 697–705

^c For babies of the same gestational age (gender-specific), birthweight below the 10th percentile and above the 90th percentile was defined as SGA and LGA, respectively, based on the INTERGROWTH-21st reference chart

^d Stillbirth was defined as fetal death at or after 20 weeks of gestation

BMI at the end of the first trimester (n=756), mean GWG z-score was -1.9 (SD=1.5); 428 (56.7%) and 327 (43.3%) had z-score within ± 2 and below -2 units of the z-score, respectively, with only one subject having a z-score above 2 units. With respect to the pregnancy outcomes, a total of 92 cases of LBW (7.5%), 195 cases of preterm birth (15.9%), 199 cases of SGA (16.2%), 134 cases of LGA (10.9%), and 47 cases of stillbirth (3.8%), were observed (Table 2).

In the main analyses, compared to the reference group with adequate GWG, women who had inadequate GWG experienced a lower risk of LGA (RR = 0.54, 95% CI: 0.36–0.80) and a higher risk of SGA (RR = 1.32, 95% CI: 0.95–1.81). For the group of excessive GWG, compared to the reference group, no significant difference in risks was observed across the outcomes that we examined, including LBW, preterm birth, SGA, or LGA (Table 3).

In the stratified analyses by BMI at the end of the first trimester, excessive GWG was associated with a higher risk of preterm birth (RR=1.59, 95% CI: 1.03–2.44) among women with normal BMI. Among women who were overweight or obese, no elevated risk was observed

with either inadequate GWG or excessive GWG compared with the reference group. In fact, a lower risk of LGA was observed in both groups of inadequate GWG (RR=0.34, 95% CI: 0.17–0.70) and excessive GWG (RR=0.45, 95% CI: 0.25–0.80). Significant statistical differences were observed between the groups with normal BMI and overweight or obesity for LGA (*p*-heterogeneity=0.01) (Table 3).

Additional analyses using the two other GWG metrics were largely consistent with the main findings. Results from the percentage GWG method suggested that, inadequate GWG was associated with higher risks of LBW (OR = 1.93, 95% CI: 1.03-3.63) and SGA (RR = 1.53, 95% CI: 1.14-2.07) and a lower risk of LGA (RR = 0.53, 95% CI: 0.38-0.77). Similar to the main analyses, overall, no significant associations were observed between excessive GWG and the outcomes examined (Additional file 1: Table S1). Among women with normal BMI at the end of the first trimester, compared to the results using the IOM classifications, results using the z-score method showed similar directions of the associations, overall (Additional file 1: Table S2).

Table 3 Association:	s between GWG by the IOM a	and adverse pregnancy outcomes	overall and stratified by BMI status

End of 1st trimester BMI	2009 IOM guidelines ^b	Pregnancy outcomes, risk ratio (95% CI) ^a			
		LBW ^c	Preterm birth ^d	SGA	LGA
	Cases (n, percent) ^e	92, 7.5%	195, 15.9%	199, 16.2%	134, 10.9%
Total (N = 1230)	Inadequate GWG (n = 517, 42.0%)	1.30 (0.67, 2.54)	0.99 (0.70, 1.40)	1.32 (0.95, 1.81)	0.54 (0.36, 0.80)
	Adequate GWG (n = 270, 22.0%)	Ref (OR = 1.00)	Ref(RR = 1.00)	Ref (RR = 1.00)	Ref(RR = 1.00)
	Excessive GWG (n = 443, 36.0%)	1.27 (0.63, 2.53)	1.22 (0.86, 1.73)	0.96 (0.67, 1.38)	0.89 (0.62, 1.29)
	Cases (n, percent)	66, 8.7%	124, 16.4%	130, 17.2%	82, 10.8%
Normal 18.5 ≤ BMI < 25 (N = 756)	Inadequate GWG (n = 343, 45.4%)	1.38 (0.63, 3.06)	1.20 (0.79, 1.84)	1.29 (0.89, 1.89)	0.65 (0.39, 1.07)
	Adequate GWG (n = 189, 25.0%)	Ref (OR = 1.00)	Ref (RR = 1.00)	Ref (RR = 1.00)	Ref (RR = 1.00)
	Excessive GWG (n = 224, 29.6%)	1.17 (0.50, 2.72)	1.59 (1.03, 2.44)	0.96 (0.61, 1.50)	1.31 (0.83, 2.06)
	Cases (n, percent)	16, 4.0%	59, 14.7%	53, 13.2%	50, 12.4%
Overweight or obese $BMI \ge 25$ (N = 402)	Inadequate GWG (n = 132, 32.8%)	0.54 (0.10, 2.85)	0.79 (0.40, 1.58)	1.10 (0.55, 2.18)	0.34 (0.17, 0.70)
	Adequate GWG (n = 65, 16.2%)	Ref (OR = 1.00)	Ref (RR = 1.00)	Ref (RR = 1.00)	Ref (RR = 1.00)
	Excessive GWG (n = 205, 51.0%)	0.87 (0.21, 3.69)	0.85 (0.46, 1.57)	0.87 (0.44, 1.72)	0.45 (0.25, 0.80)
	P-heterogeneity ^f	0.97	0.64	0.97	0.01

Institute of Medicine (IOM), gestational weight gain (GWG), body mass index (BMI), low birth weight (LBW), small for gestational age (SGA), large for gestational age (LGA), odds ratio (OR), risk ratio (RR), confidence interval (CI)

^a Multivariable model was adjusted for age (years), baseline gestational age (weeks), gestational age at delivery (weeks), BMI at 14 weeks of gestation (underweight, normal, overweight, obese), primigravida status (yes, no), treatment status (iron, placebo), marital status (married, other than married), education (0–4 years, 5–7 years, 8–11 years, ≥ 12 years), occupation (unemployed, unskilled or informal, skilled), and history of prior complications (yes, no)

^b The IOM provided recommended ranges of weekly GWG rate during the 2nd and 3rd trimesters (kg/week) by pre-pregnancy BMI status: 0.44–0.58 kg/week for underweight, 0.35–0.50 kg/week for normal weight, 0.23–0.33 kg/week for overweight, and 0.17–0.27 kg/week for obese

^c Model for estimating RR did not converge due to small number of LBW events; OR from multivariable logistic regression was reported to approximate RR instead ^d Gestational age at delivery was not adjusted in the model for preterm birth

^e Total number and percent of cases in overall and in groups of normal BMI and overweight/obesity were presented

^f P-value for heterogeneity was computed for the interaction term between GWG and BMI status at the end of 1st trimester excluding underweight women

Discussion

This study prospectively examined GWG during the second and third trimesters and adverse pregnancy outcomes in an urban pregnancy cohort with singleton births in Tanzania. We modeled GWG during the second and third trimesters of pregnancy using multiple metrics in conjunction with birth outcomes. We found that inadequate GWG was associated with a lower risk of LGA and higher risks of SGA and LBW, and excessive GWG was associated with higher risk of preterm birth, particularly among women with normal BMI.

Overall, studies have suggested heterogeneity in GWG across different SSA countries, with rising rates of excessive GWG reported in the countries of higher economic status, compared with other countries in the region [17]. In this study including healthy women in urban Tanzania, more than a third of women were found to have inadequate GWG, and a similar proportion had excessive GWG during the second and third trimesters of

pregnancy. These GWG characteristics reported are in slight contrast to those reported from studies in SSA countries. A meta-analysis by Asefa et al. recently reviewed GWG in SSA according to the IOM guidelines and found that out of the sixteen SSA studies that they examined, nine studies had more than half of pregnant women with inadequate GWG. However, four studies from lower-middle-income or middle-income countries included in the review reported higher prevalence of excessive GWG (30.6% and 32.0% from two urban studies in Cameroon, 55.5% from an urban study in South Africa, and 29.6% from a clinic-based study in South Africa) [17]. Our findings are thereby consistent with reported rising tide of excessive GWG in SSA populations with lowermiddle or middle-income status.

Prior studies in SSA have mainly focused on examining inadequate GWG given prevailing concerns for undernutrition in many countries in Africa [31]. In line with the overall literature evidence [17, 32], we reported that inadequate GWG was associated with higher risks of SGA and LBW and lower risk of LGA among pregnant women in urban Tanzania. The meta-analysis on GWG and pregnancy outcomes in SSA noted earlier reported overall significant associations between lower GWG and higher risk of LBW [17]. An association between inadequate GWG and lower risk of LGA observed in our study has also been supported in prospective studies of SSA [32] or other middle-income countries [33]. Finally, Johnson et al. prospectively examined GWG as defined by the INTERGROWTH-21st standards, and they reported that greater GWG was associated with lower risk of SGA (RR = 0.58, 95% CI: 0.46–0.72), consistent with the association found in our study [34].

While excessive GWG has been linked with higher risk of LGA and lower risks of LBW and SGA in other populations [33, 35–37], the relationships between excessive GWG and these outcomes in SSA have not been adequately examined. Limited evidence from retrospective or small-scale studies have suggested lower rate of LBW among African women with greater GWG [38-40]. A recent observational study including 170,428 pregnancies from Lebanon retrospectively examined the association between GWG and risks of SGA and LGA (percentage of SGA and LGA: 8.5 and 9.6%, respectively); the authors reported that excessive GWG was related to lower risk of SGA and higher risk of LGA across BMI categories [32]. In this study, we did not observe an association between excessive GWG and elevated risk of LGA or reduced risk of SGA or LBW among women in Tanzania. It was possible that the amount of GWG difference between excessive vs. adequate GWG groups in our study may not be sufficient enough to detect a significant difference in outcome risks. Further, with a smaller sample size compared to the Lebanon study, our study was likely underpowered. Future studies with a greater range of GWG are needed to further evaluate excessive GWG with pregnancy outcomes among African women and confirm these findings.

For the outcome of preterm birth, while no difference in the risks was observed comparing inadequate and adequate GWG groups, a higher risk was seen in the group with excessive GWG, particularly among women with normal BMI in early pregnancy. So far, literature has presented evidence on both insufficient and excessive GWGs on higher risk of preterm birth (i.e., a U-shaped relationship) [41–44]. There were a few prospective or large-scale studies that examined GWG and preterm birth among African women. One study from Malawi (n=1287) did not find significant difference in risks of preterm birth across the three GWG groups [45]. Another study among HIV-infected women in South Africa (n=471) reported higher GWG and increased risk of spontaneous preterm birth (OR=4.35, 95% CI: 1.55–12.21 for 1 kg/week increase of GWG) [46]. The earlier reported study in Lebanon also reported results supporting the U-shaped relationship between GWG and risk of preterm birth [32]. Furthermore, studies have demonstrated the association varied by pre-pregnancy BMI, with excessive GWG associated with a higher risk of preterm birth for women with greater BMI [42, 44]. Therefore, different findings across different studies may likely be due to differences in population characteristics, particularly pre-pregnancy BMI [47], different types of preterm births being examined (i.e., spontaneous vs. medically induced) [42], and failure to fully account for the correlation between GWG and gestational age [28].

GWG has long been considered as a critical marker for various in-pregnancy nutritional and physiological conditions [37, 48]. For the mechanisms of GWG and outcomes related to infant weight or size (i.e., LBW, SGA, and LGA) and prematurity, maternal weight gain reflects the health status of the mother and the growth of the fetus [13, 49]. Poor nutritional status, including macronutrient and micronutrient deficiencies, reduced immune function, and underlying maternal infection, might lead to inadequate GWG and smaller fetus growth, thus increasing the risks of LBW [50] and SGA [51]. Above factors and poor plasma volume expansion are also underlying causes for prematurity [52]. On the other hand, overnutrition [53] or impaired glucose function of the mother [54] can lead to excessive GWG, thus resulting in greater fetal growth and consequently higher risk of LGA. For the outcome of prematurity, excessive GWG may reflect underlying infection leading to increased nutrient requirement, which, if unmet, could result in preterm delivery [48]. Furthermore, studies have proposed a link between excessive GWG and higher risk of preterm birth through mechanisms related to proinflammatory response [41].

Compared to other countries, countries in SSA have long had poor rates of SGA, LBW, and prematurity, all of which have serious long-term health consequences [7, 55]. Affected newborns face neonatal and future complications, including cognitive impairment, poor growth and stunting, and noncommunicable diseases [31, 56]. On the other hand, given the rising trends in overweight and obesity, particularly among women in SSA countries experiencing transitions in economic status and dietary and nutrition transitions [20, 21], rates of excessive GWG and obesity-related pregnancy outcomes are also expected to rise. Such pregnancy complications have consequences in future obesity-related and metabolic complications for both mother and the offspring [4]. Therefore, clinical guidelines on maternal care should now evolve to start to monitor and emphasize both

inadequate and excessive GWGs in SSA countries experiencing these transitions, in effort to prevent short-term and long-term pregnancy complications. Additionally, tailoring nutrition guidelines for women based on their pregnancy BMI and focusing on both underweight and overweight in nutrition counselling for women may be beneficial. We also observed that women who had excessive GWG were more likely to be primigravida. Given the observed patterns of GWG differed by parity status (i.e., primigravida, secundigravidae), future well-powered studies are needed to examine the associations of GWG and adverse pregnancy outcomes by parity status and other maternal baseline characteristics; and to evaluate whether GWG recommendations may be adjusted according to key maternal characteristics. Lastly, given the important role of maternal diets on GWG and pregnancy outcomes and that there is likely heterogeneity in dietary patterns by region, future studies should characterize regional dietary patterns and examine GWG and pregnancy outcomes in conjunction with maternal dietary data.

Monitoring GWG is an important step for addressing inappropriate GWG and preventing its negative consequences [12, 57]. However, current evidence suggests a general lack of longitudinal monitoring system of GWG in countries of SSA [18]. Therefore, local public health practitioners should identify effective and feasible strategies integrating GWG monitoring into routine antenatal care. Further, intervention trials are needed to evaluate factors associated with optimal GWG, such as diet and physical activity [58], and aim to develop feasible programs and provide clinically useful guidelines on GWG management for pregnant women in SSA.

This present study has several strengths, including using repeated measures on pregnancy weights, characterizing GWG by multiple metrics, prospectively examining the associations between GWG and pregnancy-related outcomes with detailed covariate adjustment. Of particular importance, while past observational studies often used baseline BMI varying across study subjects when examining the associations, we adjusted and stratified the analyses by early-pregnancy BMI that was anchored at the same gestational age with the use of statistical models, leading to higher efficiency in covariate adjustment and better accuracy in stratification, compared to earlier SSA studies.

This study has some limitations. First, classification of appropriate GWG based on the IOM guidelines required the knowledge of pre-pregnancy BMI, which was not available in the study. Instead, we used BMI at the end of the first trimester to characterize GWG, which may lead to exposure misclassification. However, because BMI at the end of the first trimester was constructed as a categorical variable and the amount of GWG during the first trimester is expected to be small, exposure misclassification should be minimal and non-differential with respect to the outcomes. Secondly, since gestational age was estimated based on self-reported LMP, recall errors on the timing of LMP may lead to misclassification on any outcome that was defined based on gestational age. However, since LMP was assessed at the study baseline prior to the exposure assessment, outcome misclassification due to errors on LMP reporting would be nondifferential with respect to the exposure, thus diluting the associations. Finally, although our study was relatively large compared to other GWG studies in SSA, our stratified analyses were underpowered. Since the associations between GWG and pregnancy outcomes differ by pre-pregnancy BMI status, and women with lower prepregnancy BMI are at particularly higher risk for many pregnancy outcomes [34, 35], future studies are needed to examine the association between GWG and pregnancy-related outcomes among underweight women in SSA. Findings from the present study may be generalized to urban African women with overall good health status. Studies among diverse African populations and by regional status (i.e., urban, rural) are needed to confirm our findings.

Conclusions

Both inadequate and excessive GWG were associated with higher risks of adverse pregnancy outcomes in African women. Clinical guidelines on GWG should be developed in prevention of both inadequate and excessive GWG, particularly in SSA countries with rising trends of obesity. Intervention trials are warranted to explore effective strategies on GWG management and assess their impacts on preventing adverse pregnancy outcomes among African women.

Abbreviations

BMI: Body mass index; CI: Confidence interval; GWG: Gestational weight gain; IOM: Institute of Medicine; LBW: Low birth weight; LGA: Large for gestational age; LMP: Last menstrual period; OR: Odds ratio; SSA: Sub-Saharan Africa; SGA: Small for gestational age; RR: Risk ratio.

Supplementary Information

The online version contains supplementary material available at https://doi. org/10.1186/s12978-022-01441-7.

Additional file 1: Table S1. Associations between GWG by percentage adequacy 834 and adverse pregnancy outcomes. Table S2. Associations between GWG by z-score and adverse pregnancy outcomes among 854 women with normal BMI at 855 the end of the first trimester.

Acknowledgements

The authors would like to thank the study participants and the field teams, including study coordinators, doctors, nurses, midwives, supervisors, and the

laboratory, administrative, and data staff at Muhimbili University of Health and Allied Sciences and the clinic sites for their contributions to the study.

Author contributions

JY, WWF, DKT, JWR, MW designed analysis plan. JY conducted statistical analysis and drafted the first version of the manuscript. MW supervised statistical analysis. WWF was the principal investigator of the original trial study. All authors (JY, MW, DKT, JWR, AMD, AIA, ABP, IM, and WWF) contributed to the interpretation of the results and critical revision of the manuscript for important intellectual content. All authors read and approved the final manuscript.

Funding

The original trial study was supported by a grant from the National Institute of Child Health and Human Development (NICHD U01 HD061232). JY is supported through the doctoral program in Population Health Sciences at Harvard University. The NIH did not have any role in the design of the study, data collection, data analysis, data interpretation, or writing of this report.

Availability of data and materials

The datasets generated and/or analyzed during the current study are not publicly available due to regulatory obligations of the collaborating institutions but may be available from the corresponding author on reasonable request.

Declarations

Ethics approval and consent to participate

The Harvard School of Public Health Human Subjects Committee, the Muhimbili University of Health and Allied Sciences Research and Publications Committee, and Tanzania's National Institute for Medical Research granted institutional review board approval. Written informed consent was obtained from all women for their participation in the study.

Consent for publication

Not applicable.

Competing interests

The authors declare that they have no competing interests.

Author details

¹Department of Epidemiology, Harvard T. H. Chan School of Public Health, 677 Huntington Ave, Boston, MA 02115, USA. ²Global Center for Asian Women's Health, Yong Loo Lin School of Medicine, National University of Singapore, Singapore, Singapore. ³Department of Obstetrics and Gynaecology, Yong Loo Lin School of Medicine, National University of Singapore, Singapore, Singapore. ⁴Department of Biostatistics, Harvard T. H. Chan School of Public Health, 677 Huntington Ave, Boston, MA 02115, USA. ⁵Department of Nutrition, Harvard T. H. Chan School of Public Health, 677 Huntington Ave, Boston, MA 02115, USA. ⁶Division of Preventive Medicine, Department of Medicine, Brigham and Women's Hospital, 900 Commonwealth Avenue, Boston, MA 02115, USA. ⁷Channing Division of Network Medicine, Division of Women's Health, Department of Medicine, Brigham and Women's Hospital and Harvard Medical School, 900 Commonwealth Avenue, Boston, MA 02115, USA. ⁸Department of Global Health and Population, Harvard T.H. Chan School of Public Health, 677 Huntington Ave, Boston, MA 02115, USA. 9 Department of Obstetrics and Gynecology, School of Medicine, Muhimbili University of Health and Allied Sciences, P. O. Box 65117, Dar es Salaam, Tanzania.

Received: 25 May 2021 Accepted: 18 May 2022 Published online: 16 June 2022

References

- Kvalvik LG, Wilcox AJ, Skjærven R, Østbye T, Harmon QE. Term complications and subsequent risk of preterm birth: registry based study. BMJ. 2020;369: m1007.
- Widen EM, Whyatt RM, Hoepner LA, Ramirez-Carvey J, Oberfield SE, Hassoun A, Perera FP, Gallagher D, Rundle AG. Excessive gestational weight gain is associated with long-term body fat and weight retention at 7 y postpartum in African American and Dominican mothers with

underweight, normal, and overweight prepregnancy BMI. Am J Clin Nutr. 2015;102:1460–7.

- Rana S, Lemoine E, Granger JP, Karumanchi SA. Preeclampsia: pathophysiology, challenges, and perspectives. Circ Res. 2019;124:1094–112.
- Rich-Edwards JW, Fraser A, Lawlor DA, Catov JM. Pregnancy characteristics and women's future cardiovascular health: an underused opportunity to improve women's health? Epidemiol Rev. 2014;36:57–70.
- 5. Neiger R. Long-term effects of pregnancy complications on maternal health: a review. J Clin Med. 2017;6:76.
- Hug L, Alexander M, You D, Alkema L. for Child UI-aG: National, regional, and global levels and trends in neonatal mortality between 1990 and 2017, with scenario-based projections to 2030: a systematic analysis. Lancet Glob Health. 2019;7:e710–20.
- Katz J, Lee AC, Kozuki N, Lawn JE, Cousens S, Blencowe H, Ezzati M, Bhutta ZA, Marchant T, Willey BA, et al. Mortality risk in preterm and small-forgestational-age infants in low-income and middle-income countries: a pooled country analysis. Lancet. 2013;382:417–25.
- Karachaliou M, Georgiou V, Roumeliotaki T, Chalkiadaki G, Daraki V, Koinaki S, Dermitzaki E, Sarri K, Vassilaki M, Kogevinas M, et al. Association of trimester-specific gestational weight gain with fetal growth, offspring obesity, and cardiometabolic traits in early childhood. Am J Obstet Gynecol. 2015;212(502):e501-514.
- Blencowe H, Krasevec J, de Onis M, Black RE, An X, Stevens GA, Borghi E, Hayashi C, Estevez D, Cegolon L. National, regional, and worldwide estimates of low birthweight in 2015, with trends from 2000: a systematic analysis. Lancet Glob Health. 2019;7:e849–60.
- 10. Organization WH: UNICEF-WHO low birthweight estimates: levels and trends 2000–2015. World Health Organization; 2019.
- 11. Louise J, Deussen AR, Dodd JM: Gestational Weight Gain-Re-Examining the Current Paradigm. Nutrients. 2020; 12.
- 12. Kominiarek MA, Peaceman AM. Gestational weight gain. Am J Obstet Gynecol. 2017;217:642–51.
- 13. Weight Gain During Pregnancy: Reexamining the Guidelines. Institute of Medicine (US) and National Research Council (US) and Committee to Reexamine IOM Pregnancy Weight Guidelines. IOM; 2009.
- Gaillard R, Steegers EA, Franco OH, Hofman A, Jaddoe VW. Maternal weight gain in different periods of pregnancy and childhood cardiometabolic outcomes. The Generation R Study. Int J Obes (Lond). 2015;39:677–85.
- Durie DE, Thornburg LL, Glantz JC. Effect of second-trimester and thirdtrimester rate of gestational weight gain on maternal and neonatal outcomes. Obstet Gynecol. 2011;118:569–75.
- Zhou Y, Li H, Zhang Y, Zhang L, Liu J, Liu J. Rate of gestational weight gain and adverse pregnancy outcomes in rural nulliparous women: a prospective cohort analysis from China. Br J Nutr. 2019;122:352–9.
- Asefa F, Cummins A, Dessie Y, Hayen A, Foureur M. Gestational weight gain and its effect on birth outcomes in sub-Saharan Africa: systematic review and meta-analysis. PLoS ONE. 2020;15: e0231889.
- Wang D, Wang M, Darling AM, Perumal N, Liu E, Danaei G, Fawzi WW. Gestational weight gain in low-income and middle-income countries: a modelling analysis using nationally representative data. BMJ Glob Health. 2020. https://doi.org/10.1136/bmjgh-2020-003423.
- Coffey D. Prepregnancy body mass and weight gain during pregnancy in India and sub-Saharan Africa. Proc Natl Acad Sci U S A. 2015;112:3302–7.
- 20. Steyn NP, McHiza ZJ. Obesity and the nutrition transition in Sub-Saharan Africa. Ann N Y Acad Sci. 2014;1311:88–101.
- 21. Abrahams Z, McHiza Z, Steyn NP. Diet and mortality rates in Sub-Saharan Africa: stages in the nutrition transition. BMC Public Health. 2011;11:801.
- Symington EA, Baumgartner J, Malan L, Zandberg L, Ricci C, Smuts CM. Nutrition during pregnancy and early development (NuPED) in urban South Africa: a study protocol for a prospective cohort. BMC Pregnancy Childbirth. 2018;18:1–12.
- 23. Wrottesley SV, Pisa PT, Norris SA. The influence of maternal dietary patterns on body mass index and gestational weight gain in urban black South African women. Nutrients. 2017;9:732.
- Etheredge AJ, Premji Z, Gunaratna NS, Abioye AI, Aboud S, Duggan C, Mongi R, Meloney L, Spiegelman D, Roberts D, et al. Iron supplementation in iron-replete and nonanemic pregnant women in Tanzania: a randomized clinical trial. JAMA Pediatr. 2015;169:947–55.
- 25. Yang J, Wang D, Darling AM, Liu E, Perumal N, Fawzi WW, Wang M. Methodological approaches to imputing early-pregnancy weight based on

weight measures collected during pregnancy. BMC Med Res Methodol. 2021;21:24.

- 26. Adu-Afarwuah S, Lartey A, Okronipa H, Ashorn P, Ashorn U, Zeilani M, Arimond M, Vosti SA, Dewey KG. Maternal supplementation with smallquantity lipid-based nutrient supplements compared with multiple micronutrients, but not with iron and folic acid, reduces the prevalence of low gestational weight gain in semi-urban Ghana: a randomized controlled trial. J Nutr. 2017;147:697–705.
- Cheikh Ismail L, Bishop DC, Pang R, Ohuma EO, Kac G, Abrams B, Rasmussen K, Barros FC, Hirst JE, Lambert A, et al. Gestational weight gain standards based on women enrolled in the Fetal Growth Longitudinal Study of the INTERGROWTH-21st Project: a prospective longitudinal cohort study. BMJ. 2016;352: i555.
- Hutcheon JA, Bodnar LM, Joseph KS, Abrams B, Simhan HN, Platt RW. The bias in current measures of gestational weight gain. Paediatr Perinat Epidemiol. 2012;26:109–16.
- Villar J, Cheikh Ismail L, Victora CG, Ohuma EO, Bertino E, Altman DG, Lambert A, Papageorghiou AT, Carvalho M, Jaffer YA, et al. International standards for newborn weight, length, and head circumference by gestational age and sex: the Newborn Cross-Sectional Study of the INTERGROWTH-21st Project. Lancet. 2014;384:857–68.
- 30. Zou G. A modified poisson regression approach to prospective studies with binary data. Am J Epidemiol. 2004;159:702–6.
- Black RE, Allen LH, Bhutta ZA, Caulfield LE, de Onis M, Ezzati M, Mathers C, Rivera J. Maternal and child undernutrition: global and regional exposures and health consequences. Lancet. 2008;371:243–60.
- 32. El Rafei R, Abbas HA, Charafeddine L, Nakad P, Al Bizri A, Hamod D, Yunis KA. Association of pre-pregnancy body mass index and gestational weight gain with preterm births and fetal size: an observational study from Lebanon. Paediatr Perinat Epidemiol. 2016;30:38–45.
- 33. Li N, Liu E, Guo J, Pan L, Li B, Wang P, Liu J, Wang Y, Liu G, Baccarelli AA, et al. Maternal prepregnancy body mass index and gestational weight gain on pregnancy outcomes. PLoS ONE. 2013;8: e82310.
- Johnson W, Elmrayed SA, Sosseh F, Prentice AM, Moore SE. Preconceptional and gestational weight trajectories and risk of delivering a small-for-gestational-age baby in rural Gambia. Am J Clin Nutr. 2017;105:1474–82.
- Goldstein RF, Abell SK, Ranasinha S, Misso M, Boyle JA, Black MH, Li N, Hu G, Corrado F, Rode L, et al. Association of gestational weight gain with maternal and infant outcomes: a systematic review and meta-analysis. JAMA. 2017;317:2207–25.
- 36. Goldstein RF, Abell SK, Ranasinha S, Misso ML, Boyle JA, Harrison CL, Black MH, Li N, Hu G, Corrado F, et al. Gestational weight gain across continents and ethnicity: systematic review and meta-analysis of maternal and infant outcomes in more than one million women. BMC Med. 2018;16:153.
- Siega-Riz AM, Viswanathan M, Moos MK, Deierlein A, Mumford S, Knaack J, Thieda P, Lux LJ, Lohr KN. A systematic review of outcomes of maternal weight gain according to the Institute of Medicine recommendations: birthweight, fetal growth, and postpartum weight retention. Am J Obstet Gynecol. 2009;201(339):e331–e314.
- Nemomsa D, Mesfin F, Damena M, Asefa F. Early-pregnancy body mass index and gestational weight gain are important maternal markers of low birth weight in Harar town, Eastern Ethiopia. East Afr J Health Biomed Sci. 2018;2:1–9.
- Abubakari A, Kynast-Wolf G, Jahn A. Maternal determinants of birth weight in Northern Ghana. PLoS ONE. 2015;10: e0135641.
- 40. Tela FG, Bezabih AM, Adhanu AK. Effect of pregnancy weight gain on infant birth weight among mothers attending antenatal care from private clinics in Mekelle City, Northern Ethiopia: a facility based follow-up study. PLoS ONE. 2019;14: e0212424.
- Huang A, Ji Z, Zhao W, Hu H, Yang Q, Chen D. Rate of gestational weight gain and preterm birth in relation to prepregnancy body mass indices and trimester: a follow-up study in China. Reprod Health. 2016;13:93.
- Pigatti Silva F, Souza RT, Cecatti JG, Passini R Jr, Tedesco RP, Lajos GJ, Nomura ML, Rehder PM, Dias TZ, Oliveira PF, Silva CM. Role of body mass index and gestational weight gain on preterm birth and adverse perinatal outcomes. Sci Rep. 2019;9:13093.
- 43. McDonald SD, Han Z, Mulla S, Lutsiv O, Lee T, Beyene J, Knowledge Synthesis G, Shah P, Ohlsson A, Shah V, et al. High gestational weight gain and the risk of preterm birth and low birth weight: a systematic review and meta-analysis. J Obstet Gynaecol Can. 2011;33:1223–33.

- 44. Savitz DA, Stein CR, Siega-Riz AM, Herring AH. Gestational weight gain and birth outcome in relation to prepregnancy body mass index and ethnicity. Ann Epidemiol. 2011;21:78–85.
- 45. Gondwe A, Ashorn P, Ashorn U, Dewey KG, Maleta K, Nkhoma M, Mbotwa J, Jorgensen JM. Pre-pregnancy body mass index (BMI) and maternal gestational weight gain are positively associated with birth outcomes in rural Malawi. PLoS ONE. 2018;13: e0206035.
- 46. Madlala HP, Malaba TR, Newell ML, Myer L. Elevated body mass index during pregnancy and gestational weight gain in HIV-infected and HIVuninfected women in Cape Town, South Africa: association with adverse birth outcomes. Trop Med Int Health. 2020;25:702–13.
- 47. Santos S, Voerman E, Amiano P, Barros H, Beilin LJ, Bergström A, Charles MA, Chatzi L, Chevrier C, Chrousos GP, et al. Impact of maternal body mass index and gestational weight gain on pregnancy complications: an individual participant data meta-analysis of European North American and Australian cohorts. BJOG. 2019;126:984–95.
- Carmichael SL, Abrams B. A critical review of the relationship between gestational weight gain and preterm delivery. Obstet Gynecol. 1997;89:865–73.
- Diemert A, Lezius S, Pagenkemper M, Hansen G, Drozdowska A, Hecher K, Arck P, Zyriax BC. Maternal nutrition, inadequate gestational weight gain and birth weight: results from a prospective birth cohort. BMC Pregnancy Childbirth. 2016;16:224.
- Valero De Bernabé J, Soriano T, Albaladejo R, Juarranz M, Calle ME, Martínez D, Domínguez-Rojas V. Risk factors for low birth weight: a review. Eur J Obstet Gynecol Reprod Biol. 2004;116:3–15.
- McCowan L, Horgan RP. Risk factors for small for gestational age infants. Best Pract Res Clin Obstet Gynaecol. 2009;23:779–93.
- 52. Organization WH: Born too soon: the global action report on preterm birth. 2012.
- Ojha S, Saroha V, Symonds ME, Budge H. Excess nutrient supply in early life and its later metabolic consequences. Clin Exp Pharmacol Physiol. 2013;40:817–23.
- Deierlein AL, Siega-Riz AM, Herring A. Dietary energy density but not glycemic load is associated with gestational weight gain. Am J Clin Nutr. 2008;88:693–9.
- Zar HJ, Pellowski JA, Cohen S, Barnett W, Vanker A, Koen N, Stein DJ. Maternal health and birth outcomes in a South African birth cohort study. PLoS ONE. 2019;14: e0222399.
- Sebire NJ, Jolly M, Harris JP, Wadsworth J, Joffe M, Beard RW, Regan L, Robinson S. Maternal obesity and pregnancy outcome: a study of 287,213 pregnancies in London. Int J Obes Relat Metab Disord. 2001;25:1175–82.
- Martínez-Hortelano JA, Cavero-Redondo I, Álvarez-Bueno C, Garrido-Miguel M, Soriano-Cano A, Martínez-Vizcaíno V. Monitoring gestational weight gain and prepregnancy BMI using the 2009 IOM guidelines in the global population: a systematic review and meta-analysis. BMC Pregnancy Childbirth. 2020;20:649.
- Hill B, Skouteris H, Fuller-Tyszkiewicz M. Interventions designed to limit gestational weight gain: a systematic review of theory and meta-analysis of intervention components. Obes Rev. 2013;14:435–50.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.